Author's response to reviews

Title: Gamma-tocotrienol treatment increased peroxiredoxin-4 expression in HepG2 liver cancer cell line

Authors:

Farahani Abdul Rahman Sazli (farahsazli@gmail.com)
Zakiah Jubri (zakiah@medic.ukm.my)
Mariati Abdul Rahman (mariati_ar@dental.ukm.my)
Saiful Anuar Karsani (saiful72@um.edu.my)
Abdul Gapor Md Top (gapor@mpob.gov.my)
Wan Zurinah Wan Ngah (zurina@medic.ukm.my)

Version: 2 Date: 23 September 2013

Author's response to reviews: see over
Dear Editor,

PAPER SUBMISSION

Herein, I would like to resubmit paper entitle **Gamma-tocotrienol treatment increased peroxiredoxin-4 expression in HepG2 liver cancer cell line.** Not much differences just on the formatting of table and figure that there is overlapping when I went thorough in final PDF version. This manuscript highlighted study on the changes of protein expression in HepG2 liver cancer cell treated with gamma-tocotrienol. Gamma-tocotrienol is one of the vitamin E isomer that rich in palm oil. It is shown to have anticancer activity but the mechanism is still not clear. In this study peroxiredoxin-4 was upregulated with gamma-tocotrienol treatment in HepG2 cell and most probably a potential therapeutic target for hepatocellular carcinoma. I believe this article should be published and you will undoubtedly find that this article is in line with the BMC Complementary and Alternative Medicine policy.

Thank you.

Yours sincerely,

DR. ZAKIAH JUBRI
Corresponding author